Author’s response to reviews

Title: Deciphering risk factors for blood stream infections, bacteria species and antimicrobial resistance profiles among children under five years of age in North-western Tanzania: a multicentre study in a cascade of referral health care system

Authors:

Jeremiah Seni (senijj80@gmail.com)

Adam Mwakyoma (amwakyoma@gmail.com)

Florence Mashuda (flongoma@yahoo.co.uk)

Rehema Marando (marandorehema@gmail.com)

Maimuna Ahmed (ahmedmaimuna@yahoo.co.uk)

Rebekah DeVinney (rdevinne@ucalgary.ca)

Johann Pitout (johann.pitout@cls.ab.ca)

Stephen Mshana (stephen72mshana@gmail.com)

Version: 1 Date: 02 Jan 2019

Author’s response to reviews:

The comments raised by the editors and three reviewers were very constructive and after an extensive amendments, they have largely re-shaped our work.

Specifically, we have responded point-by-point to the comments raised and the respective amendments are presented in tabular form in the cover letter and also by the text highlight color in the respective area in the revised manuscript.